Author’s response to reviews

Title: A low-cost uterine balloon tamponade for management of postpartum hemorrhage: modeling the potential impact on maternal mortality and morbidity in sub-Saharan Africa

Authors:

Tara Herrick (therrick@path.org)
Mercy Mvundura (mmvundura@path.org)
Thomas Burke (tburke@partners.org)
Elizabeth Abu-Haydar (eabuhaydar@path.org)

Version: 2 Date: 28 Sep 2017

Author’s response to reviews:

We adjusted line 259/260 to include PPH based on Reviewer 4's feedback. We also added the author contributions to the manuscript.